3 June 2010 Ocean Veritas Status Report

Complied by Don Aurand (EM&A for BP)

A total of 4 CTD casts at four locations were made today. Weather constraints limited activities in the early afternoon as a front with high winds and heavy rains moved through the area. After a delay of approximately 2.5 hours sampling resumed and continued until 18:30 local. By that time the sea state had increased to the point that operations after dark were not safe, and so a fifth location was not occupied. 
At least twice today the R/V Gordon Gunter and the Ocean Veritas were operating in close proximity. As the number of research vessels continues to increase it seems even more advisable to establish some sort of coordination of the various programs.

The first two samples of the day, OV017 and OV018 completed the southern leg of the transect to the west of the well head which we began yesterday (see attached figure). 

No significant fluorescence was detected in either sample OV017 (approximately 5.0 km SW of the well head) or at OV018 (approximately 8.4 km SSW of the well head) also failed to produce significant fluorescence readings at any depth. The onboard LISST data also did not indicate any presence of dispersed oil sized droplets at depth, but did see some near the surface.

After completing sample OV018, we moved approximately 3.9 km to the west to begin a northerly transect. This would take us through the area where a fluorescence signal was observed on 1 June by the R/V Brooks McCall. Once on station for sample OV019 the weather delay occurred and activities ceased until 1530 local. When the sample was completed at approximately 1630 local, there was no hydrocarbon signal at any depth, which was confirmed by the preliminary LISST data.

We then moved north approximately 3.7 km. Sample OV020 (approximately 4 km W of sample OV017) was also negative for any indication of dispersed hydrocarbons, either from fluorometry or the LISST particle analyzer.
Oxygen profiles at all stations appeared typical for this time of year, with surface values around 6 to 8 mg/l, a decline to the vicinity of 4 mg/l as depth increased, followed by a rise back to approximately 6 mg/l in the colder, deep water.
Water samples for Rototox tests not were taken today since there was no indication of hydrocarbons and the rotifer hatch of yesterday yielded far fewer animals than expected. Samples taken yesterday from OV015 (processed a portion of the sampled depths) and OV016 (1160 m only), were  inoculated this morning but some of the clean water samples had to be held at 4 degrees because there are still not enough rotifers for testing. A new hatch of rotifers should be ready tomorrow morning.
Based on the lack of results today, and on information provided by Shejun Fan, Ph.D., of Fugro Global Environmental & Ocean Sciences based on ADCP data from DD3 (NTL 42916) and Discovery Enterprise (NTL 42913) that the current speed is very low and that the current flow is towards the SE below 900 m, we intend to survey the area to the SE of the well head tomorrow. The initial three proposed stations are shown in the attached figure. 

Again, any recent modeling results that might be available at the Area Command would be greatly appreciated.
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